Author's response to reviews

Title: Bone turnover in passive smoking female rat: relationships to change in bone mineral density

Authors:

Shu-guang Gao (shuguang.gao@gmail.com)
Kang-hua Li (lkh8738@gmail.com)
Mai Xu (xumai2008@163.com)
Wei Jiang (jiangwei2008@sina.com)
Hong Shen (hong.shen@gmail.com)
Wei Luo (luowei001@163.com)
Wen-shuo Xu (wenshuo2008@163.com)
Jian Tian (tianjian0228@sina.com)
Guang-hua Lei (lgh9640@gmail.com)

Version: 7 Date: 3 June 2011

Author's response to reviews: see over
Dear Editor:

Appreciations to your kind consideration for our paper submission and quick review results. We are glad to resubmit you our revised manuscript after a revision according to your professional suggestions. Meanwhile, a detailed responses describing how we have responded to the specific points raised by the reviewers are also attached by the end of this letter.

If you have any other concern regarding to our revised manuscript, please feel free to contact me. At the same time, we would like to appreciate you and the nice reviewers who had given us professional comments to improve our manuscript.

Best wishes.

Sincerely yours,

Guang-Hua Lei, Prof.
Department of Orthopaedics
Xiangya Hospital, Central South University
#87 Xiang-Ya Road
Changsha, Hunan
China, 410008
Tel: 086-731-84327326
Fax: 086-731-84327224
E-mail: lgh9640@gmail.com.
Major revisions (we require the author to make these changes)

References: Please make sure references are cited in numerical order. You have cited reference 41 on page 14, then on the same line cited 45-48, missing out 42-44. Please correct this.

Answer: The references has been corrected (page 14).

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Structure: please change 'Materials and Methods' to 'Methods'.

Answer: Done.

Tables: please remove the visible vertical lines from your tables

Answer: Done.